

Author's response to reviews

Title: Risk factors for antenatal depression, postnatal depression and parenting stress

Authors:

Bronwyn Leigh (bronwyn.leigh@austin.org.au)

Jeannette Milgrom (jeannette.milgrom@austin.org.au)

Version: 4 **Date:** 10 April 2008

Author's response to reviews: see over

9 April 2008

To the BioMed Central Editorial Production Team

Dear Sir/Madam,

Please find below formatting changes made to manuscript, as requested:

Title

p.1 Title of manuscript ‘Risk factors for antenatal depression, postnatal depression and parenting stress’ changed to sentence case

Section headings

- p. 2 Section heading ‘Abstract’ changed to sentence case.
- p.4 Section heading ‘Background’ changed to sentence case.
- p.4. Sub-heading ‘Risk factors for postnatal depression’ changed to sentence case.
- p.5 Sub-heading ‘Risk factors for antenatal depression’ changed to sentence case.
- p.5. Sub-heading ‘Parenting stress’ changed to sentence case.
- p.7 Section heading ‘Methods’ changed to sentence case.
- p.7 Sub-heading ‘Procedure and design’ changed to sentence case.
- p.8 Sub-heading ‘Antenatal screening 26-34 weeks’ changed to sentence case.
- p.8 Sub-heading ‘Demographics and psychosocial risk factors’ changed to sentence case.
- p.8,9,10 Sub-headings for scale names have **not** been changed to sentence case.
- p.9 Sub-heading ‘Antenatal risk factors battery 28-36 weeks’ changed to sentence case.
- p.10 Sub-heading ‘Postnatal measures 10-12 weeks’ changed to sentence case.
- p.11 Section heading ‘Results’ changed to sentence case.
- p.12 Sub-heading ‘Prevalence estimates’ changed to sentence case.
- p.12 Sub-heading ‘Prevalence of perinatal depression’ changed to sentence case.
- p. 12 Sub-heading ‘Prevalence of antenatal anxiety’ changed to sentence case.
- p. 12 Sub-heading ‘Primary data analyses’ changed to sentence case.
- p. 12 Sub-heading ‘Predictors of antenatal depression’ changed to sentence case.
- p.13 Sub-heading ‘Predictors of postnatal depression’ changed to sentence case.
- p. 13 Sub-heading ‘Antenatal depression as a mediator of postnatal depression’ changed to sentence case.
- p. 14 Sub-heading ‘Predictors of parenting stress’ changed to sentence case.
- p. 15 Sub-heading ‘Postnatal depression as a mediator of parenting stress’ changed to sentence case.
- p.16 Section heading ‘Discussion’ changed to sentence case.
- p. 18 Sub-heading ‘Integrative model of risk factors for antenatal depression, postnatal depression and parenting stress’ changed to sentence case.
- p. 19 Section heading ‘Conclusions’ changed to sentence case.

- p.23 Section heading ‘References’ changed to sentence case.
- p. 28 Title for Table 1 changed to sentence case.
- p. 29 Title for Table 2 changed to sentence case.
- p. 30 Title for Table 3 changed to sentence case.
- p. 31 Title for Table 4 changed to sentence case.
- p. 32 Title for Table 5 changed to sentence case.

Table and figure placement text

p. 13 ‘Insert table 1 here’ deleted

p. 13 'Insert table 2 here' deleted
p. 14 'Insert table 3 here' deleted
p. 15 'Insert table 4 here' deleted
p. 15 'Insert table 5 here' deleted
p. 18 'Insert figure 1 here' deleted

References

The list of references has been moved to before the figure legends (now p. 23-26).

Figure title

The title Figure 1.... Etc has been deleted from the image.

Tables

Tables 1 to 5 have been included with the table titles after the references and figure legend (now p. 28-32)

Additional files

This page has been deleted.

Other minor typographical changes

p.2, Background, first sentence – a comma has been inserted after “....postnatal depression is high,...” AND a comma has been inserted after “.....around 13%,.....”

p.4 The acronym (PND) has been added after the sub-heading to read: Risk factors for postnatal depression (PND)

p. 23 References:

Reference 1 has now been updated as follows: “in press” has been deleted and the journal details ‘108 (1-2), 147-157’ have been added.

Reference 9 – the initials ‘A.W.’ for the last author Gemmill have been deleted from after the surname and inserted before the surname. The ‘&’ has been replaced with ‘and’ before the last author’s name.

Reference 11 – the author’s “Cooper, P.J., Wilson, A. and Romaniuk, H.” have been deleted and replaced with ‘et al.’.

Reference 13 – a space has been inserted between “50 (5)”

p. 26 References:

Reference 54 – a comma has been deleted after the journal name Journal of Personality and Social Psychology

Reference 55 – the square symbol between 2003 and 2004 has been replaced with a hyphen.

Reference 57 – the underlining below the journal name has been removed.

Sincerely,

Professor Jeannette Milgrom,
Department of Psychology,
School of Behavioural Science,
University of Melbourne;
CEO, Parent-Infant Research Institute and
Director, Clinical and Health Psychology,
Austin Health, Melbourne, Australia